Author’s response to reviews

Title: The relationship between depression and sexual health service utilization among men who have sex with men (MSM) in Cote d'Ivoire, West Africa.

Authors:

Mark Ulanja (markulanja@gmail.com; mulanja@med.unr.edu)
Carrie Lyons (clyons8@jhmi.edu)
Sosthenes Ketende (sketende@jhu.edu)
Shauna Stahlman (sstahlml@jhu.edu)
Douada Diouf (dioufda@endatiersmonde.org)
Abo Kouame (kwagny@gmail.com)
Rebecca Ezouatchi (rbkezou@yahoo.fr)
Amara Bamba (bamba.amara@gmail.com)
Fatou Drame (amadrameh@yahoo.fr)
Benjamin Liestman (bliestml@jhu.edu)
Stefan Baral (sdbaral@hotmail.com)

Version: 2 Date: 07 Jun 2018

Author’s response to reviews:

(1) Please provide the email addresses of all authors within the title page of your manuscript

Response: Thank you for this comment. I have added the emails of all the authors.

(2) Please provide the Abstract of your study on the page following your title page, and not on the same page

Response: The abstract has been moved to a new page, separate from title page.
(3) Please provide a 'Declarations' section heading after the 'Conclusions' section of your manuscript

Response: The "Declarations" section has been added after the "conclusions" section.

(4) Please state in your 'Ethics approval and consent to participate' section whether all data used for your study is openly available or whether you received permission (and from who) to use this.

Response: Thank you for this comment. I have added this information to the heading, 'Ethics approval and consent to participate' indicating the data is not publicly available, and if needed under reasonable grounds for secondary analysis, can be obtained from the corresponding author. This will require approval by the ethical review boards in Cote DIvoire and also at the Johns Hopkins University School of Public Health.

(5) Please state at the end of your 'Authors' Contributions' section that 'All authors read and approved the manuscript'

Response: This comment has been added.

(6) Please remove your funding information from the 'Acknowledgements' section of your 'Declarations'. In this section, you should acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section. If you do not have anyone to acknowledge, please write "Not applicable" in this section.

Response: Thank you from the comment. I have made the corrections.